[role="exclude",id="redirects"]
= Deleted pages

[partintro]
--

The following pages have moved or been deleted.

--
[role="exclude",id="monitoring-xpack-kibana"]
== Configuring monitoring in {kib}

See <<monitoring-kibana>>.


[role="exclude",id="settings-xpack-kb"]
== {xpack} Settings in {kib}

include::{asciidoc-dir}/../../shared/settings.asciidoc[]

For more {kib} configuration settings, see <<settings>>.

[role="exclude",id="uptime-security"]
== Uptime security

This page has moved. Please see the new section in the {uptime-guide}/uptime-security.html[Uptime Monitoring Guide].

[role="exclude",id="tutorial-load-dataset"]
== Loading sample data

This page has moved. Please see <<tutorial-build-dashboard>>.

[role="exclude",id="creating-df-kib"]
== Creating {transforms}

This page is deleted. Please see
{stack-ov}/ecommerce-dataframes.html[Transforming the eCommerce sample data].

[role="exclude",id="ml-jobs"]
== Creating {anomaly-jobs}

This page has moved. Please see {stack-ov}/create-jobs.html[Creating {anomaly-jobs}].

[role="exclude",id="job-tips"]
== Machine learning job tips

This page has moved. Please see {stack-ov}/job-tips.html[Machine learning job tips].

[role="exclude",id="extend"]
== Extend your use case

This page was deleted. See <<xpack-graph>> and <<xpack-ml>>.


[role="exclude",id="pdf-layout-modes"]
== PDF layout modes

This page has moved. Please see <<reporting-getting-started>>.

[role="exclude",id="xpack-reporting"]
== Reporting from Kibana

This page has moved. Please see <<reporting-getting-started>>.

[role="exclude",id="add-sample-data"]
== Add sample data

This page has moved. Please see <<get-data-in>>.
